

September 12,2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 4812-review.doc).



Title: Localized Type 1 autoimmune pancreatitis superimposed upon preexisting intraductal papillary mucinous neoplasms

Author: Takahiro Urata, MD, Yoshiki Naito, MD, PhD , Yoshihiro Izumi, MD, Yukiko Takigawa, MD, Kenichi Ikebe, MD, Ginga Tonaki, MD, Hajime Iwasaki, MD, Nobuhiro Minami, MD, Rituko Yoshioka, MD,Hideki Kitada, MD,Yoshi Takekuma, MD,PhD, Hiroshi Yokomizo, MD,PhD,Michiko Nagamine, MD,Seiji Fukuda, MD,PhD, Kenji Notohara, MD,PhD, Michio Hifumi, MD, PhD

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 4812

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

Reviewed by 00020433

1. An important diagnostic criteria for autoimmune pancreatitis is elevated serum IgG4 levels which have not been provided.

→ Sorry, in this case, we have not conducted the blood test of IgG4 before the operation.

2. The patient had normal pancreatic duct as seen on MRCP and ERP. An explanation is required.

→In our case, since AIP has occur only the circumfluence of IPMNs,main pancreatic duct is normal.

3. Were there any other organ involvement.

→The IgG4 related disease was not observed in other internal organs.

Reviewed by 00045221

1. This is the interesting case and well written paper. The table is not necessary because the result is totally normal. Figure 5 can be deleted due to too many figures.

→Thank you for your suggestion. We decided to delete Table1 and Figure 5.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Takahiro Urata,MD

¹Department of Gastroenterology, Japanese Red Cross Kumamoto Hospital, Kumamoto, Japan

Address: 2-1-1, Nagamine Minami, Kumamoto city, Kumamoto, 861-8520, Japan

Phone: 096-384-2111, FAX: 096-384-3939

Mail address:tagatame0126@yahoo.co.jp